Author's response to reviews

Title: The influence of Methylphenidate on the power spectrum of ADHD children - a MEG study

Authors:

Christian Wienbruch (Christian.Wienbruch@uni-konstanz.de)
Isabella Paul (Isabella.Paul@uni-konstanz.de)
Susanne Bauer (sudebauer@surfeu.de)
Hermann Kivelitz (320062204450-0001@t-online.de)

Version: 2 Date: 23 April 2005

Author's response to reviews:

"Competing interests" section added